Author’s response to reviews

Title: Pharmacies as providers of expanded health services for people who inject drugs: A review of the legal and policy framework in six countries

Authors:

Theodore M Hammett (ted_hammett@abtassoc.com)
Son Phan (sph003@yahoo.com)
Julia Gaggin (jgaggin@fenwayhealth.org)
Patricia Case (zalafa@gmail.com)
Nicholas Zaller (N.Zaller@lifespan.org)
Alexandra Lutnick (alutnick@rti.org)
Alex H Kral (akral@rti.org)
Ekaterina Federova (ekaterina_federova@yahoo.com)
Robert Heimer (robert.heimer@yale.edu)
Will Small (wsmall@cfenet.ubc.ca)
Robin Pollini (rpollini@PIRE.org)
Leo Beletsky (lbeletsky@gmail.com)
Carl Latkin (clatkin@jhsph.edu)
Don C Des Jarlais (DDesjarlais@chpnet.org)

Version: 2 Date: 11 October 2013

Author’s response to reviews: see over
Dear Editor:

My co-authors and I are pleased to submit this revised manuscript on the legal and policy framework for expanding pharmacy-based health services for people who use drugs. We believe that the paper makes an important contribution to understanding the opportunities presented by and the challenges to increasing the availability of direct and referral health services for drug users at pharmacies, one of the most commonly accessed types of health facilities by this extremely needy and vulnerable population.

Pursuant to your request, we have revised the manuscript as follows: added “Competing Interests” and “Authors’ Contributions” sections.

We thank you for your consideration of this paper and look forward to your response.

Sincerely,

Theodore M. Hammett, Ph.D.
Vice President/Principal Associate
Abt Associates Inc.